Author’s response to reviews

Title: Effects of home-based play-assisted stimulation on developmental performances of children living in extreme poverty: a randomized single-blind controlled trial

Authors:

Berhanu Worku (brexnigussie83@yahoo.com; berhanunigussie.worku@student.uhasselt.be)
Teklu Abessa (teklugemechuabessa@student.uhasselt.be)
Mekitie Wondafrash (mwondafrash@yahoo.com)
Johan Lemmens (johan.lemmens@pxl.be)
Jan Valy (johan.valy@pxl.be)
Liesbeth Bruckers (liesbeth.bruckers@uhasselt.be)
Patrick Kolsteren (patrick.kolsteren@ugent.be)
Marita Granitzer (marita.granitzer@uhasselt.be)

Version: 2 Date: 26 Jul 2017

Author’s response to reviews:

Dear Editor-in-chief,

Language corrections are, now, made to this manuscript (following Reviewer 1 and your comments) by two independent language experts and the author of this manuscript. Moreover, the abstract in the online submission is now updated.